Author's response to reviews

Title: Repeated BCG treatment of mouse bladder selectively stimulates small GTPases and HLA antigens and inhibits single-spanning uroplakins

Authors:

Marcia R Saban (marcia-saban@ouhsc.edu)
Helen L Hellmich (hhellmic@utmb.edu)
Cindy Simpson (cindy-simpson@ouhsc.edu)
Carole A Davis (carole-davis@ouhsc.edu)
Mark L Lang (Mark-lang@ouhsc.edu)
Michael A Ihnat (michael-ihnat@ouhsc.edu)
Michael A O'Donnell (michael-odonnell@uiowa.edu)
Xue-Ru Wu (xue-ru.wu@med.nyu.edu)
Ricardo Saban (ricardo-saban@ouhsc.edu)

Version: 2 Date: 22 October 2007

Author's response to reviews:

The authors would like to thank Drs. Tsutomu Nakahara and Dah-Shyong S Yu for taking the time out their busy schedules to review this manuscripts.

We all appreciated your kind words and the suggestion to replace the term : "chronic BCG treatment" by "repeated BCG treatment". The latter was done in the title and throughout the manuscript text and figures.